Answering Reviewers

Name of Journal: World Journal of Clinical cases
Manuscript NO: 68926
Title: Establishment of risk assessment score for deep vein thrombosis after artificial liver support system treatment

Dear Editors and Reviewers:
Thank you very much for your comments regarding our manuscript. We have carefully revised the manuscripts per your suggestion.

Reviewer #1:

Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)

This study was divided into three stages to evaluate the risk factors for deep vein thrombosis (DVT) after the use of an artificial liver support system (ALSS) and establish a risk assessment score. The design of the study is reasonable, the idea is clear, and the results are detailed and reliable

A: Thank you very much for your recommendation for publication.

Sincerely.
Youwen Tan
Department of Hepatology, The Third Hospital of Zhenjiang Affiliated Jiangsu University (No.300,Daijiamen,Runzhou Distinct,Zhenjiang 212003), Zhenjiang, China
Tel: +86-511-88970796;
Fax: +86-511-88970796;
E-mail: tyw915@sina.com